Author's response to reviews

Title: "If you can't treat HPV, why test for it?" Women's attitudes to the changing face of cervical cancer prevention - a focus group study.

Authors:

Judith McRae (judemcrae@yahoo.com)
Cara Martin (cara.martin@tcd.ie)
John O'Leary (OLEARYJJ@tcd.ie)
Linda Sharp (linda.sharp@ncri.ie)

Version: 4 Date: 9 April 2014

Author's response to reviews: see over
9th April 2014

Re: “If you can’t treat HPV, why test for it?” Women’s attitudes to the changing face of cervical cancer prevention: a focus group study:

Dear Editor,

Many thanks for your consideration of our paper. I have addressed your editorial request and have provided a revised manuscript confirming that the study conforms to the RATS guidelines on qualitative research.

Please don’t hesitate to contact me if you require anything further.

Yours,

Dr Judith Murphy